

# RESEARCH ARTICLE

# Annual adult survival drives trends in Arctic-breeding shorebirds but knowledge gaps in other vital rates remain

Emily L. Weiser,<sup>1,\*,</sup> Richard B. Lanctot,<sup>2</sup> Stephen C. Brown,<sup>3</sup> H. River Gates,<sup>4</sup> Joël Bêty,<sup>5,#</sup> Megan L. Boldenow,<sup>6,#</sup> Rodney W. Brook,<sup>7,#</sup> Glen S. Brown,<sup>7,#</sup> Willow B. English,<sup>8,#</sup> Scott A. Flemming,<sup>9,10,#</sup> Samantha E. Franks,<sup>11,#</sup> H. Grant Gilchrist,<sup>12,#</sup> Marie-Andrée Giroux,<sup>13,#</sup> Andrew Johnson,<sup>14,#</sup> Steve Kendall,<sup>15,16,#</sup> Lisa V. Kennedy,<sup>9,#</sup> Laura Koloski,<sup>9,#</sup> Eunbi Kwon,<sup>17,#</sup> Jean-François Lamarre,<sup>5,#</sup> David B. Lank,<sup>18,#</sup> Christopher J. Latty,<sup>15,#</sup> Nicolas Lecomte,<sup>19,#</sup> Joseph R. Liebezeit,<sup>20,#</sup> Rebecca L. McGuire,<sup>21,#</sup> Laura McKinnon,<sup>22,#</sup> Erica Nol,<sup>9,#</sup> David Payer,<sup>15,23,#</sup> Johanna Perz,<sup>9,#</sup> Jennie Rausch,<sup>24,#</sup> Martin Robards,<sup>21,#</sup> Sarah T. Saalfeld,<sup>2,#</sup> Nathan R. Senner,<sup>25,#</sup> Paul A. Smith,<sup>12,#</sup> Mikhail Soloviev,<sup>26,#</sup> Diana Solovyeva,<sup>27,#</sup> David H. Ward,<sup>1,#</sup> Paul F. Woodard,<sup>24,#</sup> and Brett K. Sandercock<sup>28</sup>

- <sup>1</sup> U.S. Geological Survey, Alaska Science Center, Anchorage, Alaska, USA
- <sup>2</sup> U.S. Fish and Wildlife Service, Migratory Bird Management, Anchorage, Alaska, USA
- <sup>3</sup> Manomet, Inc., Manomet, Massachusetts, USA
- <sup>4</sup> National Audubon Society, Anchorage, Alaska, USA
- <sup>5</sup> Département de Biologie, Chimie et Géographie and Centre d'Études Nordiques, Université du Québec à Rimouski, Rimouski, Quebec, Canada
- <sup>6</sup> Department of Biology and Wildlife, University of Alaska Fairbanks, Fairbanks, Alaska, USA
- <sup>7</sup> Ontario Ministry of Natural Resources and Forestry, Peterborough, Ontario, Canada
- <sup>8</sup> Department of Biological Sciences, Simon Fraser University, Burnaby, British Columbia, Canada
- <sup>9</sup> Environmental and Life Sciences Graduate Program, Trent University, Peterborough, Ontario, Canada
- <sup>10</sup> Environment and Climate Change Canada, Delta, British Columbia, Canada
- <sup>11</sup> British Trust for Ornithology, Thetford, UK
- <sup>12</sup> Environment and Climate Change Canada, Ottawa, Ontario, Canada

<sup>13</sup> K.-C.-Irving Research Chair in Environmental Sciences and Sustainable Development, Université de Moncton, Moncton, New Brunswick, Canada

- <sup>14</sup> Cornell Lab of Ornithology, Cornell University, Ithaca, New York, USA
- <sup>15</sup> U.S. Fish and Wildlife Service, Arctic National Wildlife Refuge, Fairbanks, Alaska, USA
- <sup>16</sup> U.S. Fish and Wildlife Service, Hakalau Forest National Wildlife Refuge, Hilo, Hawaii, USA
- <sup>17</sup> Department of Behavioural Ecology & Evolutionary Genetics, Max Planck Institute for Ornithology, Seewiesen, Germany
- <sup>18</sup> Centre for Wildlife Ecology, Simon Fraser University, Burnaby, British Columbia, Canada
- <sup>19</sup> Canada Research Chair in Polar and Boreal Ecology, Université de Moncton, Moncton, New Brunswick, Canada
- <sup>20</sup> Audubon Society of Portland, Portland, Oregon, USA
- <sup>21</sup> Wildlife Conservation Society, Arctic Beringia Program, Fairbanks, Alaska, USA
- <sup>22</sup> Department of Multidisciplinary Studies/Biology, York University Glendon Campus, Toronto, Ontario, Canada
- <sup>23</sup> National Park Service, Alaska Regional Office, Anchorage, Alaska, USA
- <sup>24</sup> Environment and Climate Change Canada, Yellowknife, Northwest Territories, Canada
- <sup>25</sup> Department of Biological Sciences, University of South Carolina, Columbia, South Carolina, USA
- <sup>26</sup> Lomonosov Moscow State University, Moscow, Russia
- <sup>27</sup> Institute of Biological Problems of the North, Magadan, Russia
- <sup>28</sup> Norwegian Institute for Nature Research, Trondheim, Norway
- #These authors contributed equally and are listed in alphabetical order by surname.
- \*Corresponding author: eweiser@usgs.gov

Submission Date: December 18, 2019; Editorial Acceptance Date: May 4, 2020; Published June 13, 2020

# ABSTRACT

Conservation status and management priorities are often informed by population trends. Trend estimates can be derived from population surveys or models, but both methods are associated with sources of uncertainty. Many Arcticbreeding shorebirds are thought to be declining based on migration and/or overwintering population surveys, but data are lacking to estimate the trends of some shorebird species. In addition, for most species, little is known about the stage(s) at which population bottlenecks occur, such as breeding vs. nonbreeding periods. We used previously published and unpublished estimates of vital rates to develop the first large-scale population models for 6 species of Arcticbreeding shorebirds in North America, including separate estimates for 3 subspecies of Dunlin. We used the models to estimate population trends and identify life stages at which population growth may be limited. Our model for the *arcticola* subspecies of Dunlin agreed with previously published information that the subspecies is severely declining. Our results also linked the decline to the subspecies' low annual adult survival rate, thus potentially implicating factors during the nonbreeding period in the East Asian–Australasian Flyway. However, our trend estimates for all species showed high uncertainty, highlighting the need for more accurate and precise estimates of vital rates. Of the vital rates, annual adult survival had the strongest influence on population trend in all taxa. Improving the accuracy, precision, and spatial and temporal coverage of estimates of vital rates, especially annual adult survival, would improve demographic model-based estimates of population trends and help direct management to regions or seasons where birds are subject to higher mortality.

Keywords: demography, fecundity, phalarope, plover, population modeling, sandpiper, survival, waders

## LAY SUMMARY

- Documenting population trends is essential for evaluating the conservation status of wild species such as Arcticbreeding shorebirds.
- Trends can be estimated with population surveys or by predicting population growth based on survival rates and fecundity, but both methods are challenging, especially for species with large or remote geographic distributions.
- We used recent broad-scale estimates of survival and fecundity to develop population models for 6 species of Arcticbreeding shorebirds.
- The *arcticola* subspecies of Dunlin is likely in severe decline, but our trend estimates for all species showed high uncertainty.
- Uncertainty around the values of annual adult survival rates was a key driver of the uncertainty around the trend estimates.
- Our work highlights the need for better estimates of annual adult survival, seasonal survival, juvenile survival, and breeding propensity for these Arctic-breeding shorebirds.

#### La supervivencia anual de adultos impulsa las tendencias en las aves playeras que se reproducen en el Ártico, pero aún quedan vacíos de conocimiento en otras tasas vitales

#### RESUMEN

El estatus de conservación y las prioridades de manejo se derivan usualmente de las tendencias poblacionales. Las estimaciones de tendencia pueden obtenerse a partir de censos o modelos poblacionales, pero ambos métodos están asociados con fuentes de incertidumbre. Se piensa que muchas aves playeras que se reproducen en el Ártico están disminuyendo, tomando como base muestreos de poblaciones migratorias o de invernada, pero faltan datos para estimar las tendencias de algunas especies de aves playeras. Adicionalmente, para la mayoría de las especies, poco se sabe sobre las etapas en las que ocurren cuellos de botella, tales como los períodos reproductivo vs. no reproductivo. Usamos estimaciones previamente publicadas y no publicadas de tasas vitales para desarrollar los primeros modelos poblacionales de gran escala para seis especies de aves plaveras que se reproducen en el Ártico en América del Norte, incluvendo estimaciones separadas para tres subespecies de *Calidris alpina*. Usamos los modelos para estimar las tendencias poblacionales e identificar las etapas de vida en las cuales el crecimiento poblacional puede estar limitado. Nuestro modelo para la subespecie C. a. arcticola coincidió con información previamente publicada de que la subespecie está disminuyendo fuertemente. Nuestros resultados también vincularon esta disminución con la baja tasa de supervivencia anual de los adultos de la subespecie, potencialmente implicando factores durante el período no reproductivo en la ruta de vuelo de Asia Oriental-Australasia. Sin embargo, nuestras estimaciones de tendencia para todas las especies mostraron una gran incertidumbre, subrayando la necesidad de más estimaciones exactas y precisas de las tasas vitales. De las tasas vitales, la supervivencia anual de los adultos tuvo la mayor influencia en la tendencia poblacional de todos los taxones. El mejoramiento de la exactitud, la precisión y la cobertura espacial y temporal de las estimaciones de las tasas vitales, especialmente de la supervivencia anual de los adultos, mejoraría las estimaciones basadas en modelos demográficos de las tendencias poblacionales y ayudaría a orientar el manejo hacia las regiones o las estaciones donde las aves están sujetas a una mayor mortalidad.

Palabras clave: ave playera, chorlito, demografía, falaropo, fecundidad, modelo poblacional, playeros, supervivencia

# INTRODUCTION

Effective management and conservation of wildlife require knowledge of population trends. Trends can be estimated either through count-based population surveys, which measure abundance, or with demographic models, which use estimates of vital rates to predict the population growth rate. When repeated population surveys and vital rates are both available, integrated population models (IPMs) can be used to evaluate trends (Schaub and Abadi 2010). However, when survey data are too sparse to develop an IPM, vital rates can be used in a demographic model. The output can then be compared with estimates from population surveys to provide multiple lines of evidence for a population trend. Through a sensitivity or elasticity analysis (de Kroon et al. 1986, Caswell 2001), demographic models can also be used to identify which vital rates have the strongest influence on population growth rate, thus directing research and management to key life stages and relevant geographic areas.

In long-lived species, adult survival often has a strong influence on the rate of population change, while reproductive rates are more influential for short-lived species (Sæther and Bakke 2000). The relative effect of each demographic parameter on population growth or decline depends on the mean and variance of the parameter; for example, high, constant survival rates drive population growth more strongly than low or variable rates (Sæther and Bakke 2000, Wisdom et al. 2000). If population growth is limited by reproductive success, management efforts might be most effective when focused on the breeding grounds. By contrast, if adult survival has the strongest influence on the rate of change, management actions might most effectively target areas where adult survival is limited.

Identifying the limiting stage of the annual cycle is especially crucial for migratory birds, which can be affected by different factors in breeding vs. nonbreeding areas (Hostetler et al. 2015). Arctic-breeding shorebirds undertake some of the longest migrations of any birds, making nonstop flights of up to 12,000 km to spend the nonbreeding season in the tropics or Southern Hemisphere (Henningsson and Alerstam 2005, Conklin et al. 2017). Nearly half of shorebird populations worldwide have shown long-term population declines associated with anthropogenic change, but population sizes and trends are not well quantified for many species (International Wader Study Group 2003, Andres et al. 2012b, Hua et al. 2015, Smith et al. 2020). Many Arctic-breeding shorebirds use remote areas during both the breeding and nonbreeding seasons, so conducting comprehensive surveys or studies of vital rates has been logistically challenging, especially on a scale relevant to the large breeding distributions of most species (Bart and Johnston 2012).

The Arctic Shorebird Demographics Network (ASDN) monitored shorebirds at 16 field sites across Alaska, Canada, and Russia in 2008-2014 (Brown et al. 2014, Lanctot et al. 2015). The ASDN produced the first comprehensive estimates of reproductive parameters for 21 species and of adult survival for 6 species of Arctic-breeding shorebirds (Weiser et al. 2018a, 2018b). We supplemented these estimates with additional unpublished data from the ASDN and previous estimates of other demographic parameters to develop population models for 6 species of Arctic shorebirds. For each species, we estimated the rate of population change and compared our results with previous estimates of trends, which were often primarily based on population surveys in nonbreeding areas (Andres et al. 2012a, 2012b, U.S. Shorebird Conservation Plan Partnership 2016). We also quantified the elasticity value

of each vital rate to identify the demographic parameter(s) that had the strongest influence on population growth rate for each species. For influential parameters, we discuss the key gaps in knowledge that could become the focus of future research. Our study provides the first flyway-scale estimates of population trends using demographic models, providing information to prioritize future research.

#### **METHODS**

The ASDN coordinated standardized data collection at 16 field sites in Alaska, Canada, and Russia (Figure 1). Methods for collection of field data are provided in detail by Brown et al. (2014) and summarized by Weiser et al. (2018a, 2018b) and all raw data are publicly available (Lanctot et al. 2016). In the present analysis, we focus on 6 species of shorebirds for which key demographic rates, including rates of true annual adult survival corrected for emigration, have been estimated. The focal species were American Golden-Plover (Pluvialis dominica), 3 allopatric subspecies of Dunlin (Calidris alpina pacifica, C. a. arcticola, and C. a. hudsonia), Semipalmated Sandpiper (C. pusilla), Western Sandpiper (C. mauri), Red-necked Phalarope (Phalaropus lobatus), and Red Phalarope (Ph. fulicarius; Table 1). Over 95% of our data were from North American sites, so our study is primarily relevant to Nearctic-breeding populations. During migration, the arcticola subspecies of Dunlin uses the East Asian-Australasian Flyway and all of our other study populations use the 4 Americas flyways (Rodewald 2020). Where information on a particular vital rate was not available for one of our study species, we used estimates for the most closely related species; we evaluated the consequences of such uncertainty in vital rates in the population model as described below.

#### **Estimating Vital Rates**

To develop our population models, we used estimates previously derived from ASDN data from 2008 to 2014 for the mean values and variances of true annual survival rates of adults (corrected for emigration; Weiser et al. 2018b), and clutch size, daily nest survival rates, and incubation duration for each species (Weiser et al. 2018a; Table 2). For most of our study species, adult survival estimates were drawn primarily from study sites in Alaska, as sample sizes and return rates were too low at sites in eastern Canada (Figure 1). We also used published estimates of renesting propensity (Gates et al. 2013), chick survival rates (Hill 2012; other studies provided survival rates by brood, not by chick), and juvenile survival rates (Fernández et al. 2003, Rice et al. 2007, Warnock and Gill 2020; Table 2), some of which were developed at or near our study sites in previous years. All vital rates were estimated independently by previous studies over various time periods, so we did not include estimates of covariance among vital rates.



**FIGURE 1.** Locations of ASDN study sites (points) and breeding ranges (orange shading) of each species in Arctic Russia, Alaska, and Canada. Point type indicates whether data were collected for only nests or both nests and adult survival. Shapefiles for range maps were provided by BirdLife (BirdLife International and Handbook of the Birds of the World 2018). For each species, study sites are shown if we documented breeding, including some sites outside of the indicated breeding range.

We developed estimates of additional parameters for the population model from the ASDN dataset, which is publicly available (Lanctot et al. 2016). First, we estimated age of first return to the breeding grounds based on birds that we banded as chicks and later observed as adults at breeding sites (Supplementary Material Appendix A). For birds present in breeding areas, extreme weather conditions can cause >50% of females (e.g., 2 of 8 years in Gratto-Trevor 1991) or nearly all individuals (Schmidt et al. 2019) to forgo breeding. However, probability of attempting to breed is not well documented in our study species. For individuals that were present on the breeding grounds, we therefore assigned a moderately high annual nesting propensity (mean = 0.80) with moderate parameter uncertainty (standard deviation [SD] = 0.10) and interannual variation (SD<sub>vr</sub> = 0.20).

For nests that hatched at least one egg, we developed an estimate of the number of chicks hatched per nest by subtracting the species-specific mean estimate of eggs lost during incubation and the mean number of unhatched eggs per nest from the total clutch size (Weiser et al. 2018a) and assumed that all other eggs in the clutch hatched. We used a mean of 1:1 for the primary sex ratios of eggs and assumed that there was no sex bias in mortality of eggs or chicks, as there is no evidence of biased sex ratios for any of our study species (English et al. 2014, Franks et al. 2020, Hicklin and Gratto-Trevor 2020, Rubega et al. 2020, Warnock and Gill 2020).

Arctic-breeding shorebirds can renest if their first clutch fails before hatching. However, rates of renesting are not well known and have been typically underestimated, as finding and identifying renests as such is challenging (Naves et al. 2008). One experimental study of radiotracked *arcticola* Dunlin found that an average of 73% of females renested, depending on timing of failure of the clutch (Gates et al. 2013). Robust estimates were not available for our other study species, so we used the same rate of 73% across all species as the best available estimate. Renests are often expected to be less successful than initial nests due to seasonal declines in reproductive output, which are present in our study system and have been described based on the initiation date of the nest (Ruthrauff and McCaffery 2005, Hill 2012, Weiser et al. 2018a). We therefore calculated the mean difference in initiation dates between initial nests and renests for 57 documented renests in our dataset (Supplementary Material Appendix B). We used estimates of seasonal declines in breeding parameters (Ruthrauff and McCaffery 2005, Hill 2012, Weiser et al. 2018a) to evaluate how mean values of clutch size, incubation duration, daily nest survival, and chick survival changed from initial nests to renests (Table 2).

## Model Structure

We modeled each shorebird species separately with a stochastic post-breeding projection matrix model (Caswell 2001). Population models typically model only the sex that could be limiting in the system, such as the number of female young produced per adult female (Caswell 2001). Modeling a single sex provides a common denominator among species with various breeding systems. Red and Red-necked Phalaropes are polyandrous, so males are likely the limiting sex for fecundity (Liker et al. 2013, Rubega et al. 2020, Tracy et al. 2020). Our other study species show obligate biparental care of the clutch through most of the incubation period and sex ratios are generally thought to be even (Franks et al. 2020, Hicklin and Gratto-Trevor 2020, Johnson et al. 2020, Warnock and Gill 2020). For consistency, we therefore used male-based population models for all species. Female-based models for plovers and sandpipers would yield identical results for most of our study species, except that annual adult survival rates might be slightly lower for female than male Western Sandpipers (Weiser et al. 2018b).

Based on our observations of known-age breeders (Table 2), we structured the model for each species with

**TABLE 1.** Population trends of 6 species of Arctic-breeding shorebirds studied at 16 field sites in Alaska, Canada, and Russia, 2008–2014. Question marks indicate uncertainty in trend estimates, as data were often sparse.

			Current population trend		
Common name	Scientific name	Species code	Previous estimates <sup>a</sup>	This study $^{\rm b}$	
American Golden-Plover	Pluvialis dominica	AMGP	Uncertain	Uncertain	1.01 (0.47–1.32)
Dunlin <sup>c</sup>	Calidris alpina pacifica	DUNLpac	Stable	Uncertain	1.19 (0.89–1.35)
	C. a. arcticola	DUNLarc	Strong decline	Strong decline?	0.83 (0.64–1.03)
	C. a. hudsonia	DUNLhud	Stable	Uncertain	1.19 (0.88–1.35)
Semipalmated Sandpiper	C. pusilla	SESA	Stable to increase	Uncertain	1.04 (0.84–1.23)
Western Sandpiper	C. mauri	WESA	Uncertain	Increase	1.13 (0.97–1.28)
Red-necked Phalarope	Phalaropus lobatus	RNPH	Stable to decline?	Uncertain	1.08 (0.77–1.32)
Red Phalarope	Ph. fulicarius	REPH	Uncertain	Uncertain	1.15 (0.64–1.37)

<sup>a</sup> Previous estimates of short-term population trends, generally from years ~2000–2015 (Andres et al. 2012a, 2012b, U.S. Shorebird Conservation Plan Partnership 2016, Smith et al. 2020).

<sup>b</sup> Numeric values are the population growth rate ( $\lambda$ ) given as mean (95% CI).

<sup>c</sup> Three allopatric subspecies of Dunlin (Cramp and Simmons 1983, Miller et al. 2015) were modeled separately in this study.

ובטוורמוב זהז ובטובזבוור	ing uncertainty in	ו המומווברבו בז	תווומרכז' אווכוב		OUDINATIC VALUE	was used.					
Vital rate	Group	AMGP	DUNLpac	DUNLarc	DUNLhud	SESA	WESA	RNPH	REPH	Inter- annual SD	Source <sup>a</sup>
Prob. first returning to breeding site	Age 1 Age 2	1 1	0.56 (0.10) 0.28 (0.10)	0.56 (0.10) 0.28 (0.10)	0.56 (0.10) 0.28 (0.10)	0.67 (0.10) 0.26 (0.10)	0.60 (0.10) 0.33 (0.10)	0.89 (0.10) _	0.89 (0.10) -	0.02 0.02	<i></i>
	Adult <sup>b</sup>	1.00	0.16 (0.10)	0.16 (0.10)	0.16 (0.10)	0.07 (0.10)	0.07 (0.10)	0.11 (0.10)	0.11 (0.10)	0.02	-
Nesting propensity	AII	0.80 (0.10)	0.80 (0.10)	0.80 (0.10)	0.80 (0.10)	0.80 (0.10)	0.80 (0.10)	0.80 (0.10)	0.80 (0.10)	0.20	2
Prob. 4-egg clutch	Initial nests	0.94 (0.02)	0.94 (0.02)	0.96 (0.01)	0.94 (0.02)	0.90 (0.02)	0.81 (0.04)	0.90 (0.03)	0.90 (0.02)	0.02	m
	Renests	0.89 (0.04)	0.35 (0.19)	0.13 (0.06)	0.33 (0.18)	0.78 (0.05)	0.36 (0.08)	0.87 (0.03)	0.72 (0.06)	0.02	m
Prob. 3-egg clutch	Clutches with	0.79 (0.02)	0.92 (0.02)	0.90 (0.02)	0.64 (0.02)	0.84 (0.02)	0.84 (0.02)	0.83 (0.02)	0.85 (0.02)	0.02	m
Prob. 2-egg clutch	<4 eggs	0.15 (0.02)	0.08 (0.02)	0.11 (0.02)	0.29 (0.02)	0.14 (0.02)	0.15 (0.02)	0.17 (0.02)	0.10 (0.02)	0.02	mι
Prob. 1-egg clutch		0.06 (0.02)	0 (0.02)	0 (0.02)	0.07 (0.02)	0.01 (0.02)	0.02 (0.02)	0.01 (0.02)	0.05 (0.02)	0.02	τ <b>ι</b>
sex ratio or eggs	All	C.U	C.U	C.U	C.U	C.U	C.U	<b>C</b> .0	C.U		
Incubation (days)	Initial nests	26 (1)	21 (1)	21 (1)	21 (1)	19(1)	20 (1)	20 (1)	19(1)	1.00	m (
	Renests	26 (1)	21 (1)	19(1)	21 (1)	19(1)	20 (1)	20 (1)	17(1)	1.00	m,
Daily survival rate	Initial nests	0.9770	0.9870	0.9778	0.9825	0.9826	0.9776	0.9806	0.9792	0.01	m
		(0.0056)	(0.0052)	(0.0052)	(0.0048)	(0.0037)	(0.0050)	(0.0043)	(0.0045)		
	Renests	0.9557	0.7830	0.8799	0.9654	0.9844	0.9477	0.9573	0.9550	0.01	m
		(0.0146)	(0.1921)	(0.0825)	(0.0462)	(0.0040)	(0.0191)	(0.0131)	(0.0141)		
Prop. eggs hatched	AII	0.98 (0.01)	0.90 (0.01)	0.96 (0.01)	0.95 (0.01)	0.94 (0.01)	0.91 (0.01)	0.95 (0.01)	0.97 (0.01)	0.02	-
Prob. renesting	AII	0.73 (0.20)	0.73 (0.20)	0.73 (0.20)	0.73 (0.20)	0.73 (0.20)	0.73 (0.20)	0.73 (0.20)	0.73 (0.20)	0.20	4
Time between first	AII	14	20	20	20	13	16	15	15	0	-
clutch and renest											
(days)	-										ı
Chick survival	Initial nests	0.71 (0.07)	0.71 (0.07)	0.71 (0.07)	0.71 (0.07)	0.71 (0.07)	0.71 (0.07)	0.71 (0.07)	0.71 (0.07)	0.10	5
	Renests	0.23 (0.19)	0.23 (0.19)	0.23 (0.19)	0.23 (0.19)	0.23 (0.19)	0.23 (0.19)	0.23 (0.19)	0.23 (0.19)	0.10	5
Juvenile survival	All	0.44 (0.10)	0.44 (0.10)	0.44 (0.10)	0.44 (0.10)	0.44 (0.10)	0.44 (0.10)	0.44 (0.10)	0.44 (0.10)	0.05	9
Adult survival	AII	0.72 (0.33)	0.94 (0.01)	0.54 (0.08)	0.95 (0.01)	0.76 (0.09)	0.91 (0.06)	0.78 (0.15)	0.86 (0.24)	0.02	7
<sup>a</sup> Source: (1) This study Fernandez et al. (2003),	(see also Lanctot Rice et al. (2007),	et al. 2016); (2 and Warnock (	!) Gratto-Trevo et al. (1997); (7	r (1991); (3) W ) Weiser et al. (	/eiser et al. (20 (2018b).	18a); (4) Gates	et al. (2013); (	5) Hill (2012);	(6) mean of	values estir	nated by
<sup>2</sup> incluaing all ages at w dividuals delayed breec	/hich all individua ling.	als of a species	were expected	a to return to 1	the breeding c	jrounds; ages	and Z are sho	wn separatel)	/ only tor spe	ecies where	some In-

up to 4 age classes: class J = juveniles (all species), 1 = vearlings, 2 = 2-yr-olds, and 3 =all age groups in which 100% of individuals were expected to breed. For species where all individuals were expected to breed as yearlings, only classes J and 3 were included in the model; likewise, for species in which all individuals were expected to breed as 2-yr-olds, the model included only classes J, 1, and 3. Agespecific probabilities of breeding resulted in age-specific values of fecundity, but we did not vary other vital rates (including annual adult survival) among classes because insufficient data were available to develop age-specific estimates. No information on density dependence of survival or fecundity is available for our study species, so we did not include density dependence in the model. Likewise, immigration and emigration rates are not known for these species, so we assumed that emigration and immigration would be balanced, on average, at our study sites, and thus modeled each population as if it were closed.

In the model for each species, transitions among ages were described by annual adult survival (S) of each age class. Fecundity (F), the number of male juveniles produced per adult male, depended on a series of components of reproductive success. For initial nests (1), fecundity was defined as:

$$F_1 = P_a \cdot N \cdot H_1 \cdot E_1 \cdot C_1 \cdot 0.5$$

where the probability of returning to the breeding area (*P*) varied by age class (a), N = nesting propensity for birds present in the breeding area, H = probability of the nest surviving to hatch (daily survival raised to the power of incubation duration in days), E = number of eggs expected to hatch (clutch size minus number of eggs lost during incubation and number of eggs remaining unhatched in a successful nest), C = survival rate of chicks to fledging, and 0.5 = sex ratio as the proportion of eggs that were expected to be male.

Renesting (laying a second clutch) has been documented in all of our study species if the first clutch fails before hatching (Lanctot et al. 2016). In one of our study taxa (*pacifica* Dunlin), a female that successfully hatches a clutch will sometimes desert her mate and produce a new clutch with a new mate (Jamieson 2011). There is no evidence of double-brooding in the other species, and our model assumed that fecundity was male-limited, so the possibility of female Dunlin double-brooding was not relevant to our models. We therefore assumed that in our male-based model, renesting occurred only after a clutch failed before hatching. Based on previous estimates that components of fecundity are lower for renests than initial nests (Hill 2012, Gates et al. 2013) and that reproductive output declines over the season (Weiser et al. 2018a), we defined each component of fecundity separately for initial nests and renests. We defined fecundity of the renesting attempt (2) similarly to the initial nest, but conditional upon on the probability of the first nest failing and the probability of renesting (R):

$$F_2 = P_a \cdot N \cdot (1 - H_1) \cdot R \cdot H_2 \cdot E_2 \cdot C_2 \cdot 0.5$$

Total fecundity across the initial nest and renest was then taken as the sum of  $F_1$  and  $F_2$ .

Our model was stochastic, incorporating estimates of demographic variance instead of using fixed mean values to estimate population trajectories. For each vital rate, we incorporated variance among replicates based on the SD estimated by previous studies or for this study, representing uncertainty in the parameter estimates. Data on variation among years were rarely available, so we applied a relatively small interannual SD to rates that were expected to vary little among years, such as annual adult survival, and relatively larger values for components of fecundity (Table 2). We drew values from a normal distribution when appropriate, or from a beta distribution for values constrained to range from 0 to 1.

#### **Model Execution**

We used the mean values of each vital rate (Table 2) to produce a deterministic calculation of the stable age structure for each model. We used that stable structure as the starting distribution for each model. We simulated 1,000 replicates of 20 yr to fully represent interannual variation and parameter uncertainty for each species. In each replicate and year, we calculated the population size (N), values of each major vital rate (survival *S* and fecundity *F* by age class), and an estimate of stochastic elasticity (e), which indicates the relative contribution of each vital rate to population growth (de Kroon et al. 1986). We used the popbio package 2.6 (Milligan and Stubben 2007) to calculate  $\lambda$ (function "lambda"), e of major vital rates (survival and net fecundity; function "elasticity"), and *e* of lower-level vital rates (function "vitalsens") for each year and replicate. We averaged values of N, S, F, and e across years within replicates and then across replicates, and calculated the 95% confidence intervals (CIs) from the distribution of simulated values across replicates.

Given the large uncertainty around many vital-rate estimates, we then simulated additional scenarios where we reduced each vital rate by half in turn and calculated  $\lambda$  in each case. These additional scenarios explicitly demonstrate the potential implications of the uncertainty inherent in the estimates we used for many vital rates. We tested reduced vital rates in these simulations to represent worst-case scenarios in terms of population trends in these species of conservation concern.

We conducted all simulations and calculations in R 3.6.1 (R Core Team 2019). Our script to run the stochastic matrix model simulation is publicly available (Weiser 2020).

#### RESULTS

#### **Estimates of Vital Rates**

Based on the age at return of locally banded chicks (corrected for detection probability; Supplementary Material Appendix A), we estimated that in sandpipers, most individuals would return to breed in their first year (42–57%) or second year (33-36%), with the remainder (7-16%, highest in Dunlin) delaying breeding until their third year (Table 2; Supplementary Material Table S1), which broadly agreed with previous estimates (Hilden and Vuolanto 1972, Reynolds 1987, Schamel and Tracy 1991, O'Hara et al. 2005, Hicklin and Gratto-Trevor 2020, Warnock and Gill 2020). We expected 89% of Red-necked Phalaropes to return in their first year and the remaining 11% in the second year. Although numbers of returning birds banded as chicks were small (5-16 individuals per species), our estimates agreed with previous assessments with even smaller samples (Supplementary Material Appendix A). We had no information on returning American Golden-Plovers or Red Phalaropes banded as chicks and there was no previous information on age at return in those species. We therefore assumed all American Golden-Plovers returned in their first year because few are thought to spend the boreal summer in nonbreeding areas (Johnson et al. 2020), and we assumed that Red Phalaropes would show the same age at first breeding as Red-necked Phalaropes. Our models therefore contained a single adult age class for American Golden-Plovers, 2 for phalaropes, and 3 for sandpipers (Supplementary Material Table S1).

In successful nests in the ASDN dataset, 90–98% of eggs were expected to hatch for each species (Table 2). For birds observed to renest following failure of the initial clutch, the renest was initiated an average of 13-20 days after the first clutch was laid (Table 2; Supplementary Material Table S2). As per previously published estimates, adult survival rates showed some variation among species, while adult fecundity showed less variation (Figure 2). Subadult fecundity varied depending on the expected age at first breeding for each species. We used a juvenile survival rate of 0.45 (SD = 0.10, interannual SD = 0.05), which was the average from 3 previous studies (Warnock et al. 1997, Fernández et al. 2003, Rice et al. 2007) across all species due to a lack of species-specific information. The implications of the uncertainties around our vital rate estimates are detailed in the elasticity and sensitivity analyses as reported below.

#### **Model Results**

The main population models predicted that 38-45% of the post-breeding population (i.e. just before fall migration) of each species would be comprised of juveniles (Supplementary Material Table S3). Simulated population growth rates averaged near or above  $\lambda = 1.00$  (stable to increasing) for 7 out of 8 taxa (Figure 2A; Table 1), although the distributions of simulated  $\lambda$  were large in most cases (Figure 3). By contrast, *arcticola* Dunlin were expected to be declining ( $\lambda = 0.83$ ; 95% CI: 0.64–1.03), which would result in the population reaching ~3% of the current size after 20 yr in the absence of density dependence.



**FIGURE 2.** Annual population growth rate ( $\lambda$ , **A**) and transition rates (**B**, **C**) estimated by the population models for 8 taxa of shorebirds. Error bars show 95% CIs of the simulated values across 1,000 replicates. A value of 1.0 (dotted line) indicates a stable population (**A**) or the maximum possible rate of annual adult survival (**B**). Fecundity is the number of male offspring produced per breeding male per year (**C**). Values for subadult age classes (1- and 2-yr-olds) are shown only for species where breeding was delayed for some individuals. Species abbreviations are defined in Table 1.



**FIGURE 3.** Distributions of simulated population growth rates ( $\lambda$ ) across 1,000 replicates for each of 8 species and subspecies (A–H). A dashed reference line is shown at  $\lambda = 1.0$  (stable population). Species abbreviations are defined in Table 1.

Variation among taxa in population growth rates closely matched the variation in adult survival rates (Figure 2A,B). Correspondingly, elasticity values (*e*) were highest for survival rates of adults in all taxa, although juvenile survival was similarly influential for *arcticola* Dunlin (Figure 4A). In the other taxa, *e* was moderate for juvenile survival and lower for fecundity. In all taxa with multiple age classes, *e* averaged higher for fecundity of adults than subadults due to the different probabilities of breeding (Figure 4B). Among lower-level components of fecundity, the strongest effects on  $\lambda$  were from annual nesting propensity and components of the initial nesting attempt, followed by age at first breeding (Figure 5A,B). Components of a renesting attempt had the smallest elasticity values (Figure 5C).

Scenarios in which we halved each vital rate in turn provided additional evidence of the effect of each vital rate on  $\lambda$ . In all species, when adult survival was halved,  $\lambda$  was significantly lower than in the main scenario and also significantly lower than 1 (Figure 6). Halving the other vital rates did not significantly change the population growth rate, but variance was large and the change in the mean was often biologically meaningful, sometimes switching a mean estimate of population growth to decline.

# DISCUSSION

We used previously published and new estimates of vital rates to develop the first continental-scale population models for 6 species of Arctic-breeding shorebirds. Our models demonstrated the strong influence of the estimated annual adult survival rate on the predicted population trend, emphasizing the importance of accurately and precisely estimating this parameter as well as managing for conditions to maximize survival when working to prevent or mitigate population declines. Uncertainty in all parameters, especially annual adult survival, resulted in wide uncertainty around our estimated population trends, indicating the need for further information on most lifehistory stages of Arctic-breeding shorebirds.

Our models estimated stable to increasing populations for most of our study taxa, which often contradicted previous estimates. However, uncertainty was large around our trend estimates, and only the estimate for Western Sandpiper was significantly different from zero. Uncertainty around estimates of population size or trend from nonbreeding surveys is also often high (Andres et al. 2012b), so the appearance of a discrepancy between our trend estimates and those from previous studies could simply be due to chance. The uncertainty around our estimates was typically due to small sample sizes relative to the magnitude of variation inherent in the population. Variation around adult survival estimates was large partly due to difficulties in distinguishing between mortality and detectability of marked individuals. Moreover, the vital rates that we used were drawn from multiple years at multiple study sites that spanned a wide range of longitude. Thus, the uncertainty around the vital-rate estimates also included spatial and temporal heterogeneity present in the dataset.

These uncertainties highlight the need for further study of Arctic-breeding shorebirds. Study of the most influential vital rates, such as adult survival, will



**FIGURE 4.** Elasticity of population growth rate to the annual adult survival (**A**) and overall fecundity (**B**) rates of each shorebird species in each age class. Error bars indicate 95% Cls of the simulated values across 1,000 replicates. Species abbreviations are defined in Table 1.



**FIGURE 5.** Elasticity of population growth to lower-level vital rates for each species. Panels show breeding propensity (**A**), parameters for the first nest of the season (**B**), and parameters for a renesting attempt (**C**). Error bars indicate 95% CIs of elasticity values across 1,000 replicates. Species abbreviations are defined in Table 1.

be especially important for understanding population trends and any causes of decline. While annual rates of survival have been estimated for our study species (Weiser et al. 2018b), uncertainty around those estimates was large. Moreover, estimating seasonal (not just annual) survival rates would help identify when during the annual cycle these birds are most susceptible to mortality, which can then focus management actions on the most relevant periods and regions to mitigate any ongoing or expected population declines. After annual adult survival, our models indicated that juvenile survival is also a potentially important parameter in driving population trends. Juvenile survival is thus far poorly known for most Arctic-breeding shorebirds (only 3 of our study species at a small number of locations; Warnock et al. 1997, Fernández et al. 2003, Rice et al. 2007) and is difficult to evaluate given the apparently low natal site fidelity in these species, but could become easier to monitor as tracking technology continues to advance. The moderate influence of the first nest attempt on population



**FIGURE 6.** Simulated population growth rate ( $\lambda$ ) under scenarios exploring the consequences of halving each vital rate in turn. For each species or subspecies (**A**–**H**), the first point (open triangle) shows  $\lambda$  estimated by the main population models using the best estimates of vital rates (Table 2) with a dashed horizontal reference line at the mean. All other scenarios, in which the indicated parameter was reduced by half, are shown with circles. A filled circle indicates an estimate of  $\lambda$  that was significantly different from the mean value from the main model. Error bars indicate 95% CIs across 1,000 replicates. A horizontal reference line is provided at  $\lambda = 1$  (stable population; pale gray dotted line).

trend also indicates that ongoing monitoring of reproductive success is warranted and further efforts would be useful to define spatiotemporal patterns in the probability of breeding, especially if changing Arctic habitat and phenology have the potential to produce large changes in these vital rates (Galbraith et al. 2014, Senner et al. 2017, Wauchope et al. 2017, Kwon et al. 2019, Saalfeld et al. 2019).

In addition to considering the uncertainty around the estimates, comparing our trend estimates with previous work is further complicated by the possibility that the sites at which we estimated vital rates and the surveyed overwintering sites might not be equally representative of the population of interest. First, migratory connectivity is not well described for some of our study species, so vital rates measured at our breeding sites might not be directly relevant to the population counts from monitored overwintering sites. Second, in some cases, the estimates of vital rates used in our study were drawn primarily from a subset of sites, with sample sizes often much larger in Alaska than eastern Canada, and thus do not equally represent the breeding ranges of our study species. Third, siteselection bias could play a role in the estimates of trend from both breeding and overwintering areas. Study sites are often selected to maximize sample sizes of the species of interest, and thus may represent high-quality sites in years of relatively high abundance rather than representing the overall population (Fournier et al. 2019). Our breeding sites were often selected based on a combination of accessibility and bird availability, and thus might represent high-quality sites with relatively high vital rates. The same issue could apply to overwintering population surveys if monitored sites were chosen due to an initial abundance of the target species. If that initial abundance was partly due to chance, then there may appear to be a population decline over time as those sites revert to their long-term mean (Fournier et al. 2019). The potential effects of representativeness and methodology on trend estimates are an important consideration when evaluating the management needs of wild populations. When the full breeding or wintering range of a species cannot be surveyed, using multiple lines of evidence could be helpful to best define population trends.

Despite the uncertainty around our trend estimates, we note that our mean estimate of trend for *arcticola* Dunlin agreed with previous estimates that the subspecies is severely declining (Andres et al. 2012b, U.S. Shorebird Conservation Plan Partnership 2016). This subspecies shows much lower mean annual adult survival rates than our other study taxa (Weiser et al. 2018b), and our simulations highlighted the importance of this vital rate in driving population trend, suggesting that low annual adult survival is likely playing a key role in the decline of this subspecies. Our other study species have higher annual adult survival rates despite being sympatric with arcticola Dunlin on the breeding grounds, and the other subspecies of Dunlin we examined also had higher annual adult survival. Of all our study taxa, arcticola Dunlin are the only group to use the East Asian-Australasian Flyway (Gill et al. 2013). Many shorebirds in that flyway are declining, possibly as a result of habitat loss in the Yellow Sea and other crucial stopover and wintering areas which has reduced annual adult survival rates (Piersma et al. 2016, Studds et al. 2017). Our findings of a likely declining trend corresponding with low annual adult survival in arcticola Dunlin corroborate this previous evidence that reduced annual adult survival may be depressing population trends for species using this flyway.

## CONCLUSION

While our models aimed to estimate population trends for Arctic-breeding shorebirds, the uncertainty around our trend estimates highlights the need for more accurate and precise estimates of vital rates from future field studies. Despite the uncertainty, our models corroborate the evidence for a severe decline in *arcticola* Dunlin, which use the imperiled East Asian–Australasian Flyway. Our models also quantified the importance of annual adult survival in driving population trends. Improving the accuracy, precision, and spatial and temporal coverage of estimates of vital rates, especially annual or seasonal adult survival, would improve demographic model-based estimates of population trends and help direct management to regions or seasons where populations are limited.

#### SUPPLEMENTARY MATERIAL

Supplementary material is available at The Condor: *Ornithological Applications* online.

#### ACKNOWLEDGMENTS

We thank J. Lamb, B. Ross, B. Verheijen, D. Ruthrauff, the Avian Ecology Lab at Kansas State University, and 2 anonymous reviewers for comments on earlier drafts of the manuscript. We thank the many field assistants who were involved in data collection, especially field crew leaders K. Bennet, M. Burrell, S. Carvey, J. Cunningham, E. D'Astous, A. Doll, L. Pirie Dominix, T. Donnelly, S. Freeman, K. Gold, A. Gottesman, K. Grond, P. Herzog, B. Hill, D. Hodgkinson, A. J. Johnson, D. Pavlik, M. Peck, L. Pollock, S. Sapora, B. Schwarz, F. Smith, H. M. Specht, M. VanderHeyden, B. M. Walker, and B. Wilkinson. We thank local communities and landowners, including the Ukpeaġvik Iñupiat Corporation, the people of the Inuvialuit Settlement Region, Sitnasuak Native Corporation, the Kuukpik Corporation, and the North Slope Borough for permitting us to conduct research on their lands. The findings and conclusions in this article are those of the author(s) and do not necessarily represent the views of the U.S. Fish and Wildlife Service. Any use of trade names is for descriptive purposes only and does not imply endorsement by the U.S. Government.

Funding statement: Major support for the ASDN was provided by the National Fish and Wildlife Foundation (grants 2010-0061-015, 2011-0032-014, 0801.12.032731, and 0801.13.041129), the Neotropical Migratory Bird Conservation Act (grants F11AP01040, F12AP00734, and F13APO535), and the Arctic Landscape Conservation Cooperative. Additional funding for participating field sites was provided by: Alaska Department of Fish and Game, Arctic Goose Joint Venture, Arctic National Wildlife Refuge, BP Exploration (Alaska) Inc., Bureau of Land Management, Canada Fund for Innovation, Canada Research Chairs, Cape Krusenstern National Monument grant, Centre for Wildlife Ecology at Simon Fraser University, Churchill Northern Studies Centre, Cornell University Graduate School Mellon Grant, Ducks Unlimited Canada, Environment and Climate Change Canada, FQRNT (Quebec), Government of Nunavut, Indigenous and Northern Affairs Canada, Kansas State University, Kresge Foundation, Liz Claiborne and Art Ortenberg Foundation, Manomet Center for Conservation Sciences, Mississippi Flyway Council, Murie Science and Learning Center grants, National Fish and Wildlife Foundation, National Park Service, National Science Foundation (Office of Polar Programs Grant ARC-1023396 and Doctoral Dissertation Improvement Grant 1110444), Natural Resources Canada (Polar Continental Shelf Program), Natural Sciences and Engineering Research Council of Canada (Discovery Grant and Northern Supplement), Neotropical Migratory Bird Conservation Act (grant 4073), Northern Studies Training Program, Selawik National Wildlife Refuge, Trust for Mutual Understanding, Université du Québec à Rimouski, University of Alaska Fairbanks, University of Colorado Denver, University of Missouri Columbia, University of Moncton, U.S. Fish and Wildlife Service (Migratory Bird Management Division, Survey, Monitoring and Assessment Program, Alaska National Wildlife Refuge System's Challenge Cost Share Program and Avian Influenza Health and Influenza programs), U.S. Geological Survey (USGS) (Changing Arctic Ecosystem Initiative, Wildlife Program of the USGS Ecosystem Mission Area), and the W. Garfield Weston Foundation. Logistical support was provided by Arctic National Wildlife Refuge, Barrow Arctic Science Consortium, BP Exploration (Alaska) Inc., Kinross Gold Corporation, Umiaq LLC, Selawik National Wildlife Refuge (USFWS), ConocoPhillips Alaska Inc., Cape Krusenstern National Monument (National Park Service), and Sirmilik National Park (Parks Canada).

**Ethics statement:** Animal handling, marking, and monitoring procedures were approved by Environment and Climate Change Canada, Government of Nunavut, Kansas State University, National Park Service, Ontario Ministry of Natural

Resources and Forestry, University of Alaska Fairbanks, University of Moncton, U.S. Fish & Wildlife Service, and U.S. Geological Survey. All applicable international, national, and institutional guidelines for the care and use of animals were followed.

Author contributions: E.L.W. compiled the field data, designed and performed the statistical analyses, and wrote the manuscript. B.K.S. assisted with design of analyses and preparation of the manuscript. R.B.L., S.C.B., and H.R.G. led development of standardized field protocols and coordinated fieldwork. B.K.S., R.B.L., S.C.B., H.R.G., and all other authors, who are listed in alphabetical order, designed and conducted the field studies, contributed to interpreting the results, and assisted with editing the manuscript.

**Data availability:** Analyses reported in this article can be reproduced using the values in Table 2 and a publicly available R script (Weiser 2020). The raw data used to calculate the vital rates in Table 2 are also publicly available (Lanctot et al. 2016).

# LITERATURE CITED

- Andres, B. A., C. L. Gratto-Trevor, P. Hicklin, D. Mizrahi, R. I. G. Morrison, and P. A. Smith (2012a). Status of the Semipalmated Sandpiper. Waterbirds 35:146–148.
- Andres, B. A., P. A. Smith, R. I. G. Morrison, C. L. Gratto-Trevor, S. C. Brown, and C. A. Friis (2012b). Population estimates of North American shorebirds, 2012. Wader Study Group Bulletin 119:178–194.
- Bart, J., and V. Johnston (Editors) (2012). Arctic Shorebirds in North America: A Decade of Monitoring. University of California Press, Berkeley, CA, USA.
- BirdLife International and Handbook of the Birds of the World (2018). Bird species distribution maps of the world, version 2018.1. http://datazone.birdlife.org/species/requestdis
- Brown, S. C., H. R. Gates, J. R. Liebezeit, P. A. Smith, B. L. Hill, and R. B. Lanctot (2014). Arctic Shorebird Demographics Network Breeding Camp Protocol, version 5. U.S. Fish and Wildlife Service and Manomet Center for Conservation Sciences. https://arcticdata.io/catalog/#view/doi:10.18739/A2CD5M
- Caswell, H. (2001). Matrix Population Models, second edition. Sinauer Press, Sunderland, MA, USA.
- Conklin, J. R., N. R. Senner, P. F. Battley, and T. Piersma (2017). Extreme migration and the individual quality spectrum. Journal of Avian Biology 48:19–36.
- Cramp, S., and K. E. L. Simmons (1983). Handbook of the Birds of Europe, the Middle East and North Africa–The Birds of Western Palearctic, vol. 3. Oxford University Press, Oxford, UK.
- English, W. B., D. Schamel, D. M. Tracy, D. F. Westneat, and D. B. Lank (2014). Sex ratio varies with egg investment in the Red-necked Phalarope (*Phalaropus lobatus*). Behavioral Ecology and Sociobiology 68:1939–1949.
- Fernández, G., H. de la Cueva, N. Warnock, and D. B. Lank (2003). Apparent survival rates of Western Sandpiper (*Calidris mauri*) wintering in northwest Baja California, Mexico. The Auk 120:55–61.
- Fournier, A. M. V., E. R. White, and S. B. Heard (2019). Site-selection bias and apparent population declines in long-term studies. Conservation Biology 33:1370–1379.

- Franks, S., D. B. Lank, and W. H. Wilson Jr (2020). Western Sandpiper (*Calidris mauri*), version 1.0. In Birds of the World (A. F. Poole, Editor). Cornell Lab of Ornithology, Ithaca, NY, USA. https://doi. org/10.2173/bow.wessan.01
- Galbraith, H., D. W. DesRochers, S. Brown, and J. M. Reed (2014). Predicting vulnerabilities of North American shorebirds to climate change. Plos One 9:21–23.
- Gates, H. R., R. B. Lanctot, and A. N. Powell (2013). High renesting rates in Arctic-breeding Dunlin (*Calidris alpina*): A clutch-removal experiment. The Auk 130:372–380.
- Gill, R. E., C. M. Handel, and D. R. Ruthrauff (2013). Intercontinental migratory connectivity and population structuring of Dunlins from western Alaska. The Condor 115:525–534.
- Gratto-Trevor, C. L. (1991). Parental care in Semipalmated Sandpipers *Calidris pusilla*: Brood desertion by females. Ibis 133:394–399.
- Henningsson, S. S., and T. Alerstam. (2005). Barriers and distances as determinants for the evolution of bird migration links: The Arctic shorebird system. Proceedings. Biological Sciences 272:2251–2258.
- Hicklin, P., and C. L. Gratto-Trevor (2020). Semipalmated Sandpiper (*Calidris pusilla*), version 1.0. In Birds of the World (A. F. Poole, Editor). Cornell Lab of Ornithology, Ithaca, NY, USA. https://doi. org/10.2173/bow.semsan.01
- Hilden, O. I., and S. Vuolanto (1972). Breeding biology of the Rednecked Phalarope *Phalaropus lobatus* in Finland. Ornis Fennica 49:57–85.
- Hill, B. L. (2012). Factors affecting survival of Arctic-breeding Dunlin (*Calidris alpina arcticola*) adults and chicks. MSc Thesis, Department of Biology and Wildlife, University of Alaska Fairbanks, Fairbanks, AK, USA.
- Hostetler, J. A., T. S. Sillett, and P. P. Marra (2015). Full-annual-cycle population models for migratory birds. The Auk: Ornithological Advances 132:433–449.
- Hua, N., K. Tan, Y. Chen, and Z. Ma (2015). Key research issues concerning the conservation of migratory shorebirds in the Yellow Sea region. Bird Conservation International 25:38–52.
- International Wader Study Group (2003). Waders are declining worldwide. Wader Study Group Bulletin 101/102:8–12.
- Jamieson, S. E. (2011). Pacific Dunlin *Calidris alpina pacifica* show a high propensity for second clutch production. Journal of Ornithology 152:1013–1021.
- Johnson, O. W., P. G. Connors, and P. Pyle (2020). American Golden-Plover (*Pluvialis dominica*), version 1.0. In Birds of the World (P. G. Rodewald, Editor). Cornell Lab of Ornithology, Ithaca, NY, USA. https://doi.org/10.2173/bow.amgplo.01
- de Kroon, H., A. Plaisier, J. Van Groendendael, and H. Caswell (1986). Elasticity: The relative contribution of demographic parameters to population growth rate. Ecology 67:1427–1431.
- Kwon, E., E. L. Weiser, R. B. Lanctot, S. C. Brown, H. R. Gates, G. Gilchrist, S. J. Kendall, D. B. Lank, J. R. Liebezeit, L. McKinnon, et al. (2019). Geographic variation in the intensity of warming and phenological mismatch between Arctic shorebirds and invertebrates. Ecological Monographs 89:e01383.
- Lanctot, R. B., S. C. Brown, and B. K. Sandercock (2016). Data from: Arctic Shorebird Demographics Network. NSF Arctic Data Center. https://arcticdata.io/catalog/view/doi:10.18739/ A28P5V92S
- Lanctot, R. B., H. R. Gates, S. C. Brown, B. K. Sandercock, E. L. Weiser, B.K. Sandercock, and S. C. Brown (2015). 2010–2014 Final report: Using a Network of Sites to Evaluate How Climate-mediated

- Liker, A., R. P. Freckleton, and T. Székely (2013). The evolution of sex roles in birds is related to adult sex ratio. Nature Communications 4:1587.
- Miller, M. P., S. M. Haig, T. D. Mullins, L. Ruan, B. Casler, A. Dondua, H. R. Gates, J. M. Johnson, S. Kendall, P. S. Tomkovich, et al. (2015). Intercontinental genetic structure and gene flow in Dunlin (*Calidris alpina*), a potential vector of avian influenza. Evolutionary Applications 8:149–171.
- Milligan, B. G., and C. J. Stubben (2007). Estimating and analyzing demographic models using the popbio package in R. Journal of Statistical Software 22:11.
- Naves, L. C., R. B. Lanctot, A. R. Taylor, and N. P. Coutsoubos (2008). How often do Arctic shorebirds lay replacement clutches? Wader Study Group Bulletin 115:2–9.
- O'Hara, P. D., G. Fernández, F. Becerril, H. de la Cueva, and D. B. Lank (2005). Life history varies with migratory distance in Western Sandpipers *Calidris mauri*. Journal of Avian Biology 36:191–202.
- Piersma, T., T. Lok, Y. Chen, C. J. Hassell, H. Y. Yang, A. Boyle, M. Slaymaker, Y. C. Chan, D. S. Melville, Z. W. Zhang, and Z. Ma (2016). Simultaneous declines in summer survival of three shorebird species signals a flyway at risk. Journal of Applied Ecology 53:479–490.
- R Core Team (2019). R: A Language and Environment for Statistical Computing. R Foundation for Statistical Computing, Vienna, Austria. https://www.R-project.org/
- Reynolds, J. D. (1987). Mating system and nesting biology of the Red-necked Phalarope *Phalaropus lobatus*: What constrains polyandry? Ibis 129:225–242.
- Rice, S. M., J. A. Collazo, M. W. Alldredge, B. A. Harrington, and A. R. Lewis (2007). Local annual survival and seasonal residency rates of Semipalmated Sandpipers (*Calidris pusilla*) in Puerto Rico. The Auk 124:1397–1406.
- Rodewald, P. (Editor) (2020). The Birds of the World Online. Cornell Laboratory of Ornithology, Ithaca, NY, USA. https:// birdsoftheworld.org/bow/home
- Rubega, M. A., D. Schamel, and D. M. Tracy (2020). Red-necked Phalarope (*Phalaropus lobatus*), version 1.0. In Birds of the World (S. M. Billerman, Editor). Cornell Lab of Ornithology, Ithaca, NY, USA. https://doi.org/10.2173/bow.renpha.01
- Ruthrauff, D. R., and B. J. McCaffery (2005). Survival of Western Sandpiper broods on the Yukon-Kuskokwim Delta, Alaska. The Condor 107:597–604.
- Saalfeld, S. T., D. C. McEwen, D. C. Kesler, M. G. Butler, J. A. Cunningham, A. C. Doll, W. B. English, D. E. Gerik, K. Grond, P. Herzog, et al. (2019). Phenological mismatch in Arcticbreeding shorebirds: Impact of snowmelt and unpredictable weather conditions on food availability and chick growth. Ecology and Evolution 9:6693–6707.
- Sæther, B.-E., and O. Bakke (2000). Avian life history variation and contribution of demographic traits to the population growth rate. Ecology 81:642–653.

- Schamel, D., and D. M. Tracy (1991). Breeding site fidelity and natal philopatry in the sex role-reversed Red and Red-necked Phalaropes. Journal of Field Ornithology 62:390–398.
- Schaub, M., and F. Abadi (2010). Integrated population models: A novel analysis framework for deeper insights into population dynamics. Journal of Ornithology 152:227–237.
- Schmidt, N. M., J. Reneerkens, J. H. Christensen, M. Olesen, and T. Roslin (2019). An ecosystem-wide reproductive failure with more snow in the Arctic. PLoS Biology 17:e3000392.
- Senner, N. R., M. Stager, and B. K. Sandercock (2017). Ecological mismatches are moderated by local conditions for two populations of a long-distance migratory bird. Oikos 126:61–72.
- Smith, P. A., L. McKinnon, H. Meltofte, R. B. Lanctot, A. D. Fox, J. O. Leafloor, M. Soloviev, A. Franke, K. Falk, M. Golovatin, et al. (2020). Status and trends of tundra birds across the circumpolar Arctic. Ambio 49:732–748.
- Studds, C. E., B. E. Kendall, N. J. Murray, H. B. Wilson, D. I. Rogers, R. S. Clemens, K. Gosbell, C. J. Hassell, R. Jessop, D. S. Melville, et al. (2017). Rapid population decline in migratory shorebirds relying on Yellow Sea tidal mudflats as stopover sites. Nature Communications 8:14895.
- Tracy, D. M., D. Schamel, and J. Dale (2020). Red Phalarope (*Phalaropus fulicarius*), version 1.0. In Birds of the World (S. M. Billerman, Editor). Cornell Lab of Ornithology, Ithaca, NY, USA. https://doi.org/10.2173/bow.redpha1.01
- U.S. Shorebird Conservation Plan Partnership (2016). Shorebirds of Conservation Concern in the United States of America—2016. http://www.shorebirdplan.org/science/ assessment-conservation-status-shorebirds/
- Warnock, N. D., and R. E. Gill (2020). Dunlin (*Calidris alpina*), version 1.0. In Birds of the World (S. M. Billerman, Editor). Cornell Lab of Ornithology, Ithaca, NY, USA. https://doi.org/10.2173/bow.dunlin.01
- Warnock, N., G. W. Page, and B. K. Sandercock (1997). Local survival of Dunlin wintering in California. The Condor 99:906–915.
- Wauchope, H. S., J. D. Shaw, Ø. Varpe, E. G. Lappo, D. Boertmann, R. B. Lanctot, and R. A. Fuller (2017). Rapid climate-driven loss of breeding habitat for Arctic migratory birds. Global Change Biology 23:1085–1094.
- Weiser, E. L. (2020). Arctic shorebird population model: U.S. Geological Survey software release. https://doi.org/10.5066/P9DZZ1OB
- Weiser, E. L., S. C. Brown, R. B. Lanctot, H. R. Gates, K. F. Abraham, R. L. Bentzen, J. Bêty, M. L. Boldenow, R. W. Brook, T. F. Donnelly, et al. (2018a). Life-history tradeoffs revealed by seasonal declines in reproductive traits of Arctic-breeding shorebirds. Journal of Avian Biology 49:jav01531.
- Weiser, E. L., R. B. Lanctot, S. C. Brown, H. R. Gates, R. L. Bentzen, J. Bêty, M. L. Boldenow, W. B. English, S. E. Franks, L. Koloski, et al. (2018b). Environmental and ecological conditions at Arctic breeding sites have limited effects on true survival rates of adult shorebirds. The Auk: Ornithological Advances 135:29–43.
- Wisdom, M. J., L. S. Mills, and D. F. Doak (2000). Life stage simulation analysis: Estimating vital-rate effects on population growth for conservation. Ecology 81:628–641.